

# **ORIGINAL REPORT**

# SERIAL ANALYSIS OF CARDIOPULMONARY FITNESS AND ECHOCARDIOGRAPHY IN PATIENTS WITH FABRY DISEASE UNDERGOING ENZYME REPLACEMENT THERAPY

Sheng-Hui TUAN, MD<sup>1,2</sup>, Pao-Chin CHIU, MD<sup>3</sup>, I-hsiu LIOU, MD<sup>4</sup>, Wen-Hsien LU, MD, PhD<sup>3</sup>, Hung-Ya HUANG, MD<sup>4</sup>, Shin-Yi WU, MD<sup>4</sup>, Guan-bo CHEN, MD<sup>5</sup> and Ko-Long LIN, MD<sup>4,6</sup>

<sup>1</sup>Department of Rehabilitation Medicine, Cishan Hospital, Ministry of Health and Welfare, Kaohsiung, Taiwan

<sup>2</sup>Department of Physical Therapy, Shu-Zen Junior College of Medicine and Management, Kaohsiung, Taiwan

<sup>3</sup>Department of Pediatrics, Kaohsiung Veterans General Hospital, Kaohsiung, Taiwan

<sup>4</sup>Department of Physical Medicine and Rehabilitation, Kaohsiung Veterans General Hospital, Kaohsiung, Taiwan

<sup>5</sup>Department of Internal Medicine, Kaohsiung Armed Forces General Hospital, Kaohsiung, Taiwan <sup>6</sup>Department of Physical Therapy, Fooyin University, Kaohsiung, Taiwan

Objective: Fabry disease, a rare X-linked disorder, can lead to exercise intolerance. In Taiwan, the cardiac variant of Fabry disease has a significantly higher prevalence than the classic variant. The cardiac variant of Fabry disease primarily involves the heart. Enzyme replacement therapy has been used to treat both variants. We aimed to study the impact of enzyme replacement therapy on exercise and cardiac structures between the classic (CL-FD) and cardiac variant (CV-FD) Fabry disease.

Design and methods: Retrospective analysis of 2 groups of patients with Fabry disease (5 patients with the classic variant and 5 with the cardiac variant), who were undergoing enzyme replacement therapy. Patients were assessed annually for 3 years using symptom-limited cycle ergometry and echocardiography.

Results: Subjects were 5 women, mean age 53 (standard deviation (SD) 14.05) years with CL-FD Fabry disease, and 5 men, mean age 65 (SD 2.35) years with CV-FD. The percentage of peak oxygen consumption to predicted value for all included patients was significantly lower (78.78% (SD 12.72)) than 100%. Annual serial measurement showed that peak metabolic equivalent and percentage of peak oxygen consumptiondecreased significantly over a period of 3 years in patients with CV-FD (p = 0.002, and p = 0.004, respectively), but not in those with CL-FD. There were no significant changes in annual serial measurements of left ventricular mass or interventricular septal thickness in patients with either variant of Fabry disease over a period of 3 years.

Conclusion: Peak exercise capacity of the patients with Fabry disease was lower than that of normal peers. Peak exercise capacity decreased over time

# LAY ABSTRACT

Patients with classic or cardiac variants of Fabry disease have lower exercise capacity than their healthy peers. It is recommended that patients commence enzyme replacement therapy at an early stage in order to improve clinical outcomes. This study examined the effects of enzyme replacement therapy on exercise capacity and the structure of the heart in 10 patients with Fabry disease who were undergoing enzyme replacement therapy. Measurements were taken each year for 3 consecutive years. No significant differences were found in the structure of the heart (left ventricular mass or interventricular septal thickness) between patients with these 2 types of Fabry disease. Peak exercise capacity decreased over time in patients with cardiac variant Fabry disease, but remained the same in patients with classic variant Fabry disease

# in patients withCV-FD, but remained the same in patients with CL-FD.

Key words: Fabry disease; enzyme replacement therapy; cardiopulmonary exercise testing; peak oxygen consumption.

Accepted Jan 16, 2020; Published Feb 27, 2020

JRM-CC 2020; 3: 1000028

Correspondence address: Ko-Long Lin, Department of Physical Medicine and Rehabilitation, Kaohsiung Veterans General Hospital, No. 386, Ta-Chung 1st Road, Kaohsiung 813, Taiwan. E-mail: kllinvghks@gmail.com

abry disease (FD) is an X-linked disorder that results  $\Gamma$  in deficiency of the lysosomal enzyme  $\alpha$ -galactosidase

(1). This enzyme deficiency leads to accumulation of glycosphingolipid, primarily globotriaosylceramide (Gb3), which is deposited chiefly in endothelial and vascular smooth-muscle cells (2). In hemizygous males, FD may present as early as 5 years of age, with characteristic features, such as neuropathic pain, hypohidrosis or anhidrosis, disseminated angiokeratoma, cornea verticillata, and microalbuminuria. Over time, male patients with FD develop renal failure, cardiac dysfunction and stroke. This disease phenotype is currently known as classic FD (CL-FD) (3). As many as two-thirds of female carriers may also have clinically significant manifestations of FD, although with a more variable and attenuated phenotype (4). Life expectancy is reduced in both males and females, but this is more apparent in males.

Two variants of FD that have manifestations primarily involving the heart (5, 6) or kidneys (7) have been reported in the last 2 decades. Patients with the cardiac variant (CV-FD) lack the classical symptoms of FD and present in the 5<sup>th</sup> to 8<sup>th</sup> decades of life with left ventricular hypertrophy (LVH), arrhythmias, and/or cardiomyopathy (8). CL-FD is very rare; the estimated birth prevalence rate in Western countries is approximately 1/40,000 (9). However, an unexpectedly high prevalence of the CV-FD mutation IVS4+919G=>A (IVS4) among both newborns and patients with idiopathic hypertrophic cardiomyopathy (1 in 1,500–1,600 males, 1 in 750–800 females) has been recognized in the Taiwan Chinese population (10, 11).

Patients with FD have decreased exercise tolerance and often report fatigue, even from childhood. The aetiology of this phenomenon is not fully understood. Cardiac and pulmonary involvement had been reported in patients with FD. Up to 36% of male patients with FD with obstructive ventilator defect were found to have pulmonary involvement. Decreased diffusing capacity, haemoptysis, alveolar precipitates and lamellar inclusion bodies have also been described among patients with FD (12, 13). LVH has been shown to occur in as many as 37% of patients with FD and the hypertrophy appears to be related to alteration of the left ventricular structure (14).

Enzyme replacement therapy (ERT) with either agalsidase-alpha or agalsidase-beta has been developed for treatment of FD. ERT may decrease cardiac mass and reduce renal accumulations of Gb3 (15). Trials comparing ERT with placebo show significant improvement with ERT with regard to microvascular endothelial deposits of Gb3 and pain-related quality of life (16). Based on these insights, early initiation of ERT has been advocated. However, to date, no studies have shown convincing clinical evidence for the superiority of agalsidase-alpha or agalsidase-beta (17). The long-term influence of ERT on the risk of morbidity and mortality related to FD is also unknown (16).

FD has multi-factorial effects on the cardiovascular system. Exercise capacity can be used as an integrated measure of cardiovascular function and allows the effects of treatment to be monitored. However, most



research has focused on electrocardiographic and echocardiographic parameters. Only a few studies have used cardiopulmonary exercise testing (CPET) to assess patients with FD. One study found decreased exercise tolerance and a precipitous reduction in diastolic blood pressure (DBP) at higher levels of exercise during CPET in patients with FD. It concluded that the significant decrease in DBP in these patients may explain deficits in exercise tolerance (18). Another study found that male patients with FD were unable to attain predicted maximal heart rate during exercise or to achieve normal exercise levels (19). A few small cohort studies using cycle ergometry in CPET also showed that ERT was associated with a small improvement in anaerobic threshold (AT) (19) and exercise tolerance in patients with FD (20).

To the best of our knowledge, no studies have used direct CPET to compare exercise capacity between patients with CL-FD and CL-FD who are undergoing ERT. Since the prevalence of CV-FD is unexpectedly high in Taiwan, we aimed to assess the impact of ERT on exercise tolerance by serial noninvasive CPET in each variant of FD overtime.

# **METHODS**

This retrospective cohort study was performed at a tertiary medical centre in Taiwan

The study was approved by the Institutional Review Board of the Kaohsiung Veterans General Hospital (number: VGHKS17-CT11-11). Informed consent was obtained from each patient before the study.

#### Subject characteristics

Inclusion criteria were: patients with FD confirmed by a documented pathogenic mutation in the  $\alpha$ -galactosidase A gene, who received regular follow-up with complete transthoracic echocardiographic examination at the cardiology outpatient clinic. The patients were undergoing routine ERT with agalsidase- $\alpha$  (Replagal<sup>TM</sup>, Shire Human Genetic Therapies (HGT), Inc., Cambridge, MA, USA), 0.2 mg/kg every other week, and were referred to the rehabilitation department for serial CPET from June 2015 to October 2017. Informed consent was obtained from each patient during each CPET session.

Exclusion criteria were: significant coronary artery disease; significant arrhythmia; concurrent known pulmonary disease; and age less than 18 years.

Patients' basic characteristics were recorded, including sex, age, body weight, height, plasma Gb3 level, and plasma lyso-Gb3 level. The study was approved by the Institutional Review Board of Kaohsiung Veterans General Hospital (number VGHKS17-CT11-11).

#### Cardiopulmonary exercise test

Symptom-limited cycle ergometry was used to measure the subjects' exercise capacity. The testing system comprised a bicycle, a flow module, a gas analyser, and an electrocardiographic monitor (Metamax 3B, Cortex Biophysik GmbH Co., Germany). The bicycle stress test was performed under the supervision of an experienced physiatrist (K.L.L). Patients pedalled for 1 min at zero resistance, after which the bicycle ergometer gradually



increased the resistance at a ramp rate of 10 W/min. The test was stopped when the patients demonstrated subjective unbearable symptoms (21). Oxygen consumption (VO<sub>2</sub>) and carbon dioxide production (VCO<sub>2</sub>) during the exercise test were measured by the breath-by-breath method. In addition, metabolic equivalent (MET), minute ventilation (VE), blood pressure (BP), and heart rate (HR) were measured throughout the exercise test. The AT was determined by the VE/VO<sub>2</sub> and VE/VCO<sub>2</sub> methods (22). VO, peak was the maximum oxygen uptake measured at peak exercise.VO, peak to predicted value (peakVO,%) was the percentage compared with the normal standards for cardiopulmonary responses to exercise using a cycle ergometer test among Taiwan adults (23). The change in peakVO<sub>2</sub> in the serial 3 years was defined as  $\Delta peakVO2\% = 100\% \times [(peakVO2 of the 2nd$ or the 3rd year) - (peakVO2 of the first year)]/ (peakVO2 of the first year). Comparison of measured and predicted ApeakVO<sub>2</sub>% was performed, with the predicted VO<sub>2</sub> calculated using 2 reference equations. One of the reference equations was the normal standards for CPET using a cycle ergometer test in Taiwan adults (23). The equation was developed for normal standards for  $VO_{2max}$  from a US national registry. The equation was:  $VO2max (mlkg^{-1}min^{-1}) = 45.2 - 0.35 * Age - 10.9 * Sex (male = 1;)$ female=2) - 0.15\*Weight (pounds) + 0.68\*Height (inches) -0.46\*Exercise Mode (treadmill=1; bike=2) (R=0.79, R<sup>2</sup>=0.62, standard error of the estimate =  $6.6 \,\mathrm{ml \, kg^{-1} \, min^{-1}}$ ). It could be applied to exercise testing using cyclemeter and treadmill. The equation was determined using multiple regression analysis, and comparisons were made with the widely-used Wasserman and European equations, which showed it performed better than traditional equations, particularly among women and individuals  $\geq$ 60 years old.

## Pulmonary function test

Pulmonary function testing was performed by spirometry at rest. Forced vital capacity (FVC) and forced expiratory volume in 1 s (FEV1) were measured.

# Echocardiography

Echocardiographic testing was undertaken according to local protocol and carried out annually in all patients. One experienced cardiologist measured left ventricular end-diastolic diameter (LVEDD), left atrial dimension (LAD), interventricular

septal thickness at end-diastole (IVS), and left ventricular posterior wall thickness at end-diastole (LVPW) according to the guidelines of the American Society of Echocardiography and the European Society of Cardiology. Left ventricular (LV) mass was calculated according to the Devereux formula, as follows: LV mass= $[1.04 \times ((LVEDD + IVS + LVPW)^3 - (LVEDD)^3) -$ 13.6] (25). LV mass index (LVMI) (g/m<sup>2</sup>) was calculated by dividing LV mass by body surface area (BSA) (26) to define the presence of LVH (LVMI >134/110 g/m<sup>2</sup> for men/women).

#### Statistical analysis

SPSS for Windows version 19.0 (2010 release, IBM Corp., Armonk, NY, USA) was used for all analyses. Data were expressed as means (standard deviation; SD). Normality and homoscedasticity were checked before each analysis. The  $\chi^2$  test was used to test for differences in distribution between categorized variables. Mann–Whitney *U* test was used for non-normally distributed variables for comparison between patients with CL-FD and those with CV-FD. Differences in data for CPET and echocardiography among the first, second, and third years of evaluation were analysed by repeated measures 1-way analysis of variance (ANOVA) and a Bonferroni *post hoc* test. A *p*-value <0.05 was considered statistically significant.

#### **RESULTS**

#### Patient characteristics

Twelve patients met the study inclusion criteria, but 2 dropped out for personal reasons (moving home to other county, and receiving ERT at a different hospital). Therefore, a final total of 10 subjects were included in the study (5 females, mean age 53 years (SD 14.05) in the CL-FD group and 5 males, mean age 65 years (SD 2.35) in the CV-FD group). Mean duration from initiation of ERT to the first year of evaluation for this study was 75.30 months (SD 34.85) in the CL-FD group and 29.70 months (SD 29.09) in the CV-FD group. Six patients were diagnosed with LVH by LVMI; 2 in the CL-FD group, and 4 in the CV-FD group. There were no significant differences between patients with CL-FD and those with CV-FD

Characteristics	All (n = 10)	CL-FD group $(n = 5)$	CV-FD group $(n=5)$	<i>p</i> -value <sup>a</sup>
Age, years, mean (SD)	59.00 (11.41)	53.00 (14.05)	65.00 (2.35)	0.151
Height, cm, mean (SD)	159.72 (1.30)	153.62 (1.47)	167.35 (0.40)	0.111
Weight, kg, mean (SD)	58.43 (9.79)	52.06 (5.93)	66.40 (7.50)	0.032*
BMI, kg/m <sup>2</sup> , mean (SD)	23.01 (3.67)	22.51 (4.91)	23.64 (1.71)	0.413
Resting SBP, mmHg, mean (SD)	119.67 (16.58)	112.40 (7.40)	128.75 (21.50)	0.286
Resting DBP, mmHg, mean (SD)	72.22 (15.62)	64.40 (14.52)	82.00 (11.83)	0.111
Resting HR, bpm, mean (SD)	66.22 (11.04)	69.00 (12.10)	62.75 (10.05)	0.556
FVC, L, mean (SD)	2.51 (0.59)	2.22 (0.55)	2.88 (0.44)	0.111
FVCP, %, mean (SD)	83.33 (8.70)	85.40 (10.29)	80.75 (6.70)	0.413
FEV1, L, mean (SD)	1.95 (0.43)	1.87 (0.53)	2.06 (0.32)	0.730
FEV1P, %, mean (SD)	80.67 (11.90)	86.60 (9.91)	73.25 (10.69)	0.190
<sup>b</sup> Duration of ERT use, mean (SD)	52.50 (38.65)	75.30 (34.85)	29.70 (29.09)	0.222
Number of patients with LVH, n, %	6 (60)	2 (40)	4 (80)	0.197

\**p* < 0.05.

<sup>a</sup>Refers to the *p*-value of a Mann–Whitney U test for continuous variables and  $\chi^2$  test for non-continuous variables between patients with classic Fabry disease and cardiac variant Fabry disease.

<sup>b</sup>Duration of ERT from first use to the time receiving baseline exercise testing.

CL-FD: classic Fabry disease; CV-FD: cardiac variant Fabry disease; BMI: body mass index; SBP: systolic blood pressure; DBP: diastolic blood pressure; ERT: enzyme replacement therapy; HR: heart rate; FVC: functional vital capacity; FVCP: percentage of predicted forced vital capacity; FEV1: force expiratory volume at 1 min; FEV1P: percentage of predicted forced expiratory volume at 1 min; LVH: left ventricular hypertrophy; SD: standard deviation.

Table II. Descriptive data for peakVO,% for all study subjects with Fabry disease (FD) at first, second, and third years of evaluation

Year of evaluation	Patients	peakVO <sub>2</sub> , % Mean (SD)	95% CI
1	Total	80.10 (17.16)	70.33-90.20
	CL-FD group	77.02 (14.45)	64.23-83.27
	CV-FD group	85.97 (21.16)	60.57-96.12
2	Total	76.89 (14.72)	66.54-84.06
	CL-FD group	77.98 (12.97)	68.62-86.18
	CV-FD group	78.98 (18.83)	58.25-94.85
3	Total	78.78 (12.72)	71.29-82.65
	CL-FD group	84.75 (5.02)	80.93-87.91
	CV-FD group	73.51 (18.15)	52.61-87.89

peakVO<sub>2</sub>%: percentage of measured peak oxygen consumption to predicted value: 95% CI: 95% confidence interval: CL-FD: classic Fabry disease: CV-FD: cardiac variant Fabry disease; SD: standard deviation.

in patient characteristics and at the first year of CPET testing pulmonary function data (FVC and FEV1) at the first evaluation year, except for body weight. Males were significantly heavier (66.40 kg (SD 7.50) than females (52.06 kg (SD 5.93)) (p=0.032), but no significant difference was noted in BMI (Table I).

# Cardiorespiratory exercise testing and echocardiography

Mean peakVO<sub>2</sub>% for all study participants with FD in the serial 3 years were all significantly lower than 100%. The 95% confidence interval (95% CI) of peakO<sub>2</sub>% of each group was also lower than 100% (Table II).



There were no significant differences in MET at the point of AT (AT MET), Peak power output (W), peak SBP. peak DBP. peak HR. or peak HRR in the serial 3 vears in either of the 2 groups over time CL-VD and CV-FD groups. Peak metabolic equivalent (peak MET) and peakVO<sub>2</sub>% decreased significantly in the CV-FD group (p=0.002, p=0.004, respectively), but not in the CL-FD group in serial measurements over 3 years. In the CV-FD group, post hoc analysis showed significant differences in peak MET (p=0.028) and peak VO<sub>2</sub>% (p=0.036) between the first and third years of evaluation. Peak respiratoru exchangs ratio (RER) in the serial 3 years showed significant difference in the CL-FD group. Post hoc analysis found that peak RER in the third year was significantly higher than in the first year (p=0.005) (Table III).

Although a trend of gradually increasing IVS and LVMI was found in the CV-FD group, no significant differences in IVS, LVMI, or LVM in the serial 3 years were noted in the CL-VD or CV-FD groups (Table III).

# Percentage differences between years for CL-FD and CV-FD groups

The percentage difference between the third and first years for peak MET, peakVO, and peakVO,% were all significantly lower in the CV-FD group than in the CL-FD group (all p=0.016), but no significant differences

Test	Group	First year Mean (SD)	Second year Mean (SD)	Third year Mean (SD)	F value	<i>p</i> -value <sup>a</sup>
AT MET	CL-FD	3.28 (0.63)	3.10 (0.58)	3.34 (0.66)	0.638	0.553
	CV-FD	4.60 (1.48)	4.18 (1.38)	3.95 (1.34)	6.202	0.088
PEAK MET	CL-FD	4.78 (1.22)	4.72 (1.03)	5.08 (0.97)	4.73	0.640
	CV-FD	6.55 (1.70)	5.93 (1.50)	5.43 (1.43) <sup>b</sup>	21.279	0.002*
Peak VO <sub>2</sub>	CL-FD	0.88 (0.30)	0.85 (0.22)	0.94 (0.26)	0.595	0.574
-	CV-FD	1.55 (0.53)	1.39 (0.48)	1.26 (0.49)	12.955	0.007*
Peak power output, W	CL-FD	59.40 (19.71)	58.00 (17.56)	63.40 (16.15)	0.316	0.613
	CV-FD	106.00 (45.13)	89.00 (22.45)	91.25 (30.90)	0.424	0.673
Peak SBP	CL-FD	134.40 (7.80)	140.60 (23.66)	147.20 (15.42)	0.791	0.486
	CV-FD	151.75 (17.80)	153.25 (11.62)	159.75 (33.18)	0.312	0.743
Peak DBP	CL-FD	74.20 (5.22)	75.60 (4.10)	79.00 (9.82)	0.616	0.524
	CV-FD	74.75 (2.06)	74.25 (1.89)	77.50 (13.18)	0.253	0.652
Peak HR	CL-FD	129.00 (15.17)	109.4 (30.44)	108.0 (29.91)	1.674	0.247
	CV-FD	118.50 (12.29)	121.75 (21.61)	113.75 (16.82)	0.443	0.662
Peak RER	CL-FD	1.02 (0.04)	1.04 (0.04)	1.10 (0.02) <sup>b</sup>	9.416	0.008*
	CV-FD	1.11 (0.02)	1.09 (0.03)	1.10 (0.02)	0.820	0.484
Peak HRR	CL-FD	24.80 (14.41)	18.6 (13.58)	12.40 (14.60)	2.559	0.138
	CV-FD	25.75 (13.72)	23.75 (11.56)	18.75 (16.09)	0.264	0.777
peakVO <sub>2</sub> %	CL-FD	77.02 (14.45)	77.98 (12.97)	84.75 (5.02)	1.040	0.397
	CV-FD	85.97 (21.16)	78.98 (18.83)	73.51 (18.15) <sup>b</sup>	16.547	0.004*
IVS (mm)	CL-FD	11.00 (2.00)	10.80 (2.28)	13.42 (4.85)	3.202	0.137
	CV-FD	13.00 (4.08)	13.25 (3.95)	13.73 (3.73)	0.422	0.674
LVMI	CL-FD	152.27 (51.59)	139.14 (30.23)	169.29 (64.11)	1.394	0.302
	CV-FD	169.42 (56.94)	178.82 (72.97)	178.41 (70.84)	0.158	0.857
LVM (g)	CL-FD	228.98 (84.76)	205.15 (43.92)	250.59 (96.11)	1.581	0.264
	CV-FD	294.16 (77.98)	308.11 (104.61)	305.35 (105.58)	0.101	0.905

Table III. Cardiorespiratory exercise testing and echocardiography results at first, second, and third years of evaluation

\*p<0.05.

<sup>a</sup>Within-group difference using repeated measures 1-way analysis of variance (ANOVA).

<sup>b</sup>Post hoc analysis using Bonferroni test showed *p*-value <0.05 compared with first year. CL-FD: classic Fabry disease; CV-FD: cardiac variant Fabry disease; AT: anaerobic threshold; MET: metabolic equivalent; AT MET: MET at the point of AT; peak MET: MET at peak exercise; peak VO,: peak oxygen consumption; peak HR: heart rate at peak exercise; peak RER: respiratory exchange ratio at peak exercise; HRR at 1 min: heart rate reserve at 1 min after termination of the test; peakVO,%; percentage of measured peak oxygen consumption to predicted value; IVS: interventricular septal thickness at end-diastole; LVMI: left ventricular mass index; LVM: mass of left ventricle; SD: standard deviation.



#### Cardiac effects of ERT in patients with Fabry disease p. 5 of 8

Table IV. Peak exercise capacity and echocardiography differences between second and first years, and third and first years

Measurement	Group	Difference between second and first years Mean (SD)	<i>p</i> -value <sup>a</sup>	Difference between third and first years Mean (SD)	<i>p</i> -value <sup>a</sup>
∆Peak MET (%)	CL-FD	1.83 (27.90)	> 0.99	8.96 (23.08)	0.016*
	CV-FD	-9.06 (6.12)		-17.07 (3.74)	
∆peakVO <sub>2</sub> (%)	CL-FD	0.81 (27.95)	0.905	9.84 (27.88)	0.016*
CV-FD CL-FD No CL-FD No CV-FD No CV-FD No CV-FD No	CV-FD	-10.03 (8.03)		-19.31 (6.68)	
	CL-FD Norm 1 <sup>d</sup>	-2.09 (6.48)	0.825 <sup>b</sup>	-5.34 (7.90)	0.319 <sup>b</sup>
	CL-FD Norm 2 <sup>d</sup>	-1.31 (0.27)	0.873 <sup>b</sup>	-2.63 (0.53)	0.372 <sup>b</sup>
	CV-FD Norm 1 <sup>d</sup>	0.53 (2.10)	0.122 <sup>c</sup>	0.67 (1.19)	0.015* <sup>c</sup>
	CV-FD Norm 2 <sup>d</sup>	-2.20 (0.11)	0.146 <sup>c</sup>	-4.40 (0.22)	0.020* <sup>c</sup>
∆peakVO <sub>2</sub> % (%)	CL-FD	4.36 (27.37)	0.905	13.36 (23.32)	0.016*
	CV-FD	-7.65 (6.22)		-14.35 (3.78)	
∆LVM (%)	CL-FD	-4.37 (23.91)	0.730	11.41 (21.80)	0.905
	CV-FD	3.55 (8.81)		3.97 (28.86)	
∆LVMI (%)	CL-FD	-3.32 (25.27)	0.730	12.52 (25.00)	0.730
	CV-FD	4.00 (9.08)		5.55 (29.18)	

\*p<0.05.

<sup>a</sup>Comparison between CL-FD and CV-FD groups by Mann–Whitney *U* test except for <sup>b</sup>and <sup>c,b</sup>Comparison between CL-FD and CL-FD-norm by Mann–Whitney *U* test.

<sup>d</sup>Norm 1, predicted based on the reference equation from FRIEND registry (24) and Norm 2, predicted based on the reference equation from Taiwan study (23).  $\Delta$  (%): percentage of the data for the second or third year minus the first year divided by the first year; peak MET: metabolic equivalent at peak exercise; CL-FD: classic Fabry disease; CV-FD: cardiac variant Fabry disease; peakVO<sub>2</sub>; peak oxygen consumption; peakVO<sub>2</sub>%; percentage of measured peak oxygen consumption to predicted value; LVMI: left ventricular mass index; LVM: mass of left ventricle; SD: standard deviation.

were noted in LVM and LVMI. There were no significant differences in any CPET and echocardiography data between the second and first years (**Table IV**).

# Difference between measured and predicted values of peak oxygen consumption (Table IV)

In comparisons between the second and first years, no significant differences were found in either CL-FD or CV-FD groups between actual and predicted values (Table IV). In comparisons between the third and the first years, there was no significant difference in the actual values we measured for the CV-FD group were significantly lower than predicted values (p=0.015 and p=0.020, respectively).

# DISCUSSION

To the best of our knowledge this is the first study to compare exercise capacity in patients with CL-FD and CV-FD who are undergoing regular ERT. The main finding of this study was that peak exercise capacity in patients with FD (whether CV-FD or CL-FD) who were undergoing ERT was lower than in their normal peers. Peak exercise capacity decreased over time in patients with CV-FD undergoing ERT, but remained the same in patients with CL-FD undergoing ERT.

There was a significant difference in body weight between the 2 groups, due to the sex of the subjects. Significant positive correlation has been shown between oxygen consumption and body weight in both cycle ergometry and treadmill exercise (27, 28). Therefore, peak MET was chosen to indicate cardiopulmonary fitness in this study. MET is the objective measure of the ratio of the rate at which a person expends energy, relative to their body weight, set by convention at 3.5 ml of oxygen per kg per min (29). Given that peak MET is the peak oxygen consumption relative to body weight, this was used to eliminate the potential bias that derived from body weight.

A number of studies have proven the efficacy of ERT in treatment of cardiac abnormalities in patients with FD. The proven long-term effects of ERT on cardiomyopathy in FD include a significant reduction in left ventricular mass (30, 31), improvement in myocardial function (30, 31), reduction in the accumulation of Gb3 in the heart (16), a lower level of cardiovascular complications (32), and improvement in quality of life (16). However, few studies have evaluated the clinical efficacy of ERT on cardiopulmonary function. In 2006, Bierer et al. (20)conducted a small cohort study, and performed serial CPETs at baseline and every 3 months over a period of at least 18 months. They found that exercise tolerance (VO2 max and oxygen pulse) increased in patients with FD (mean age 32 years) who were receiving ERT. However, there were only 4 patients with FD in their study (20).

Lobo et al. (19) used bicycle stress testing to measure the exercise capacity of patients with FD (mean age 42.8 (SD 11) years). They found that exercise capacity was reduced in patients with FD compared with that predicted from normative population data, which was similar to our result. They also found mild improvement in AT in the first year of ERT, but no consistent further increase beyond the first year (19).

There was no detailed description of the type of FD in patients recruited to these previous studies. Since CV-FD is also called "late-onset FD", and usually presents in the 5th to 8th decades of life (8), most of the subjects in these studies might have had CL-FD. The mean age of the patients with CV-FD in our study was 65 years (SD 2.35), which was consistent with previous studies. Weidemann et al. (30) found that patients with FD (mean age 42 years (SD 7)) undergoing ERT had higher exercise capacity obtained by bicycle stress exercise than those without ERT 3 years after the initiation of ERT (106 W (SD 14) and 122 W (SD 26), respectively; p=0.014) (30). Our results showed that, in patients with CL-FD, AT MET, peak MET, peak O2, Peak power output (W), and peakVO2% were higher in the third year than in the first year of evaluation, but there was no significant difference in the above parameters, except for peak RER (p=0.008), on repeated measures ANOVA. In contrast, in patients with CV-FD, peak exercise capacity gradually became significantly worse in serial measurements over the 3 years. Given that there was a difference in percentage peak oxygen consumption in the serial 3 years between measured and predicted values of peakVO, for the CV-FD group, the reduction in peakVO, in CV-FD was due not only to the physiological changes that occur during normal ageing, but also to the cardiac involvement of the disease.

Cardiac involvement in both CL-FD and CV-FD begins with hypertrophy, which leads to hypertrophy cardiomyopathy and development of heart failure (33). Of interest, and with no current explanation, patients with CV-FD tend to have mutation-specific cardiac involvement (34). There have been no studies comparing the difference in cardiac involvement between patients with CL-FD and those with CV-FD. One study has shown that significant cardiomyocyte substrate accumulation in patients with IVS4 leads to severe and irreversible cardiac fibrosis prior to the development of LVH or other significant cardiac manifestations (35). In our study, 80% of subjects in the CV-FD group had LVH, and it is possible that they might already have had cardiac fibrosis, which could explain the different efficacy of ERT on peak exercise capacity and LVM between the CL-FD group (in which only 40% of subjects had LVH) and the CV-FD group. This result was also similar to that of Weidemann et al.'s study (30), which found that ERT resulted in a significant reduction in LVM, and a higher exercise capacity in patients with FD without cardiac fibrosis, while patients with FD who did have cardiac fibrosis showed a minor reduction in LVH and no improvement in exercise capacity with ERT (30).

A high RER indicates that predominantly carbohydrates are being metabolized. If peak RER exceeds 1.0, additional energy is supplied anaerobically, during which induced metabolic acidosis occurs (36). Patients with heart failure generally have difficulty reaching RER  $\geq 1.0$ during exercise tolerance testing (37, 38). In our study the mean peak RER in both groups at the 3 follow-ups were all > 1.0. Thus, it is possible that none of the patients had heart failure, even though LVH was noted on echocardiography in both groups. We also observed that, in the first 2 years, mean peak RER in the CL-FD group was lower than in the CV-FD group. Since all the participants in the CL-FD group were women, this difference could be due sex differences in substrate oxidation during aerobic exercise, which results in men having a higher RER than women (39, 40). There was also a significant increase in



peak RER at the third year in the CL-FD group, while it remained the same in the CV-FD group. It is possible that ERT is more beneficial for patients with CL-FD than for those with CV-FD.

Only a few studies have examined the exercise capacity or exercise prescription of patients with FD. Many of these studies found that the lower exercise capacity in patients with FD might result from cardiac abnormalities (19, 20, 41). Refraining from physical exercise has been shown to result in exercise intolerance, in parallel with the evolution of FD (42). Exercise intolerance in patients with FD may be multi-factorial. In 2016, a pilot study by Schmitz et al. (19) showed that a 12-month strength/ circuit exercise training protocol (3 training sessions per week, 30–45 min per session, 90–135 min per week) may help to improve endurance capacity, muscle strength and overall well-being in patients with FD. In addition to ERT, we consider that adequate exercise training is important for patients with FD to ameliorate their exercise capacity.

CL-FD occurs predominantly in males, although some heterozygous females have a more severe phenotype that resembles CL-FD (3, 4). Studies have shown that clinical manifestations in heterozygous females, partly due to skewed X-chromosome inactivation, can vary widely, from no apparent clinical disease to full expression of CL-FD (43).

In Taiwan, CV-FD is the most common form of FD (11). Patients with CV-FD are generally asymptomatic for most of their lives and present in the 5<sup>th</sup> to 8<sup>th</sup> decade with LVH, hypertrophic cardiomyopathy, conduction abnormalities, and arrhythmias (8, 14). CL-FD occurs predominantly in males, although some heterozygous females have a more severe phenotype that resembles CL-FD (3, 4). Studies have shown that clinical anifestations in heterozygous females, partly due to skewed X-chromosome inactivation, can vary widely, from no apparent clinical disease to full expression of CL-FD (43).

In our study it was surprising that all of the CL-FD group were female and all of the CV-FD group male. The authors had reviewed all the medical records, including alpha-galactosidase A enzyme activity, cardiac nuclear magnetic resonance imaging, and echocardiography, and confirming the diagnosis of each variants in our participants were correct. However, it was difficult for the authors to explain the difference of the sex distribution in our participants.

# Study limitations

The current study has some limitations. First, there were only 5 patients in each group. Although FD is a relatively rare disease worldwide, small numbers of subjects might be more difficult to get a statistically significant difference. Secondly, all of the subjects were recruited from a single medical centre in southern Taiwan and the results might be generalizable only to similar populations. Thirdly, all the patients with FD in this study were already receiving ERT before the first year of evaluation, and



there was no baseline data for their exercise capacity and LVMI prior to commencing ERT. Although the patient profiles for the first year of evaluation in both groups were not significantly different, and there was significant difference in not only the absolute values of peak exercise capacity, but also the relative values compared with the first year of evaluation (as shown in Table IV), the causal relationship between ERT and peak exercise capacity could not be determined directly. Fourthly, although there was no significant difference in the percentage of patients with LVH and LVMI in each group, we did not assess the status of cardiac fibrosis, a factor that has been proven to influence the efficacy of ERT in patients with FD.

# Conclusion

This study was the first to compare exercise capacity, measured by direct CPET, and LVMI between patients with CL-FD and those with CV-FD who were undergoing ERT in annual follow-up for 3 consecutive years. The results showed that peak exercise capacity in patients with FD who were undergoing ERT was lower than that of normal peers. This study also found that peak exercise capacity decreased over a period of 3 years in patients with CV-FD who were undergoing ERT, but remained the same in patients with CL-FD who were undergoing ERT.

Larger prospective studies, using more sophisticated imaging, e.g. cardiac magnetic resonance imaging (MRI), and measurement of Gb3 levels, are warranted in order to elucidate the difference in efficacy of ERT in patients with CL-FD and those with CV-FD with regard to cardiac structures and exercise capacity.

The authors have no conflicts of interest to declare

# REFERENCES

- 1. Eng CM, Desnick RJ. Molecular basis of Fabry disease: mutations and polymorphisms in the human alpha-galactosidase A gene. Hum Mutat 1994; 3: 103–111.
- Brown LK, Miller A, Bhuptani A, Sloane MF, Zimmerman MI, Schilero G, et al. Pulmonary involvement in Fabry disease. Am J Respir Crit Care Med 1997; 155: 1004–1010.
- MacDermot KD, Holmes A, Miners AH. Anderson-Fabry disease: clinical manifestations and impact of disease in a cohort of 98 hemizygous males. J Med Genet 2001; 38: 750–760.
- MacDermot KD, Holmes A, Miners AH. Anderson-Fabry disease: clinical manifestations and impact of disease in a cohort of 60 obligate carrier females. J Med Genet 2001; 38: 769–775.
- Nakao S, Takenaka T, Maeda M, Kodama C, Tanaka A, Tahara M, et al. An atypical variant of Fabry's disease in men with left ventricular hypertrophy. N Engl J Med 1995; 333: 288–293.
- Ommen SR, Nishimura RA, Edwards WD. Fabry disease: a mimic for obstructive hypertrophic cardiomyopathy? Heart 2003; 89: 929–930.
- Kotanko P, Kramar R, Devrnja D, Paschke E, Voigtlander T, Auinger M, et al. Results of a nationwide screening for Anderson-Fabry disease among dialysis patients. J Am Soc Nephrol 2004; 15: 1323–1329.
- 8. Sachdev B, Takenaka T, Teraguchi H, Tei C, Lee P, McKenna WJ, et al. Prevalence of Anderson-Fabry disease in male

patients with late onset hypertrophic cardiomyopathy. Circulation 2002; 105: 1407–1411.

- Meikle PJ, Hopwood JJ, Clague AE, Carey WF. Prevalence of lysosomal storage disorders. Jama 1999; 281: 249–254.
- Chien YH, Lee NC, Chiang SC, Desnick RJ, Hwu WL. Fabry disease: incidence of the common later-onset alphagalactosidase A IVS4+919G-->A mutation in Taiwanese newborns – superiority of DNA-based to enzyme-based newborn screening for common mutations. Mol Med 2012; 18: 780–784.
- 11. Lin HY, Chong KW, Hsu JH, Yu HC, Shih CC, Huang CH, et al. High incidence of the cardiac variant of Fabry disease revealed by newborn screening in the Taiwan Chinese population. Circ Cardiovasc Genet 2009; 2: 450–456.
- Rosenberg DM, Ferrans VJ, Fulmer JD, Line BR, Barranger JA, Brady RO, et al. Chronic airflow obstruction in Fabry's disease. Am J Med 1980; 68: 898–905.
- Kelly MM, Leigh R, McKenzie R, Kamada D, Ramsdale EH, Hargreave FE. Induced sputum examination: diagnosis of pulmonary involvement in Fabry's disease. Thorax 2000; 55: 720–721.
- Linhart A, Palecek T, Bultas J, Ferguson JJ, Hrudova J, Karetova D, et al. New insights in cardiac structural changes in patients with Fabry's disease. Am Heart J 2000; 139: 1101–1108.
- Rombach SM, Smid BE, Linthorst GE, Dijkgraaf MG, Hollak CE. Natural course of Fabry disease and the effectiveness of enzyme replacement therapy: a systematic review and meta-analysis: effectiveness of ERT in different disease stages. J Inherit Metab Dis 2014; 37: 341–352.
- El Dib R, Gomaa H, Carvalho RP, Camargo SE, Bazan R, Barretti P, et al. Enzyme replacement therapy for Anderson-Fabry disease. Cochrane Database Syst Rev 2016; 7: Cd006663.
- Vedder AC, Linthorst GE, Houge G, Groener JE, Ormel EE, Bouma BJ, et al. Treatment of Fabry disease: outcome of a comparative trial with agalsidase alfa or beta at a dose of 0.2 mg/kg. PloS One 2007; 2: e598.
- Bierer G, Kamangar N, Balfe D, Wilcox WR, Mosenifar Z. Cardiopulmonary exercise testing in Fabry disease. Respiration 2005; 72: 504–511.
- Lobo T, Morgan J, Bjorksten A, Nicholls K, Grigg L, Centra E, et al. Cardiovascular testing in Fabry disease: exercise capacity reduction, chronotropic incompetence and improved anaerobic threshold after enzyme replacement. Intern Med J 2008; 38: 407–414.
- Bierer G, Balfe D, Wilcox WR, Mosenifar Z. Improvement in serial cardiopulmonary exercise testing following enzyme replacement therapy in Fabry disease. J Inherit Metab Dis 2006; 29: 572–579.
- Medicine ACoS. ACSM's Guidelines for exercise testing and prescription. 9th edn. Philadelphia, PA: Lippincott Williams & Wilkins, 2013.
- Washington RL. Cardiorespiratory testing: anaerobic threshold/respiratory threshold. Pediatr Cardiol 1999; 20: 12–15; discussion 16.
- Hsi WL, Lan C, Lai JS. Normal standards for cardiopulmonary responses to exercise using a cycle ergometer test. J Formos Med Assoc 1998; 97: 315–322.
- 24. de Souza ESCG, Kaminsky LA, Arena R, Christle JW, Araujo CGS, Lima RM, et al. A reference equation for maximal aerobic power for treadmill and cycle ergometer exercise testing: analysis from the FRIEND registry. Eur J Prev Cardiol 2018: 2047487318763958.
- 25. Devereux RB, Reichek N. Echocardiographic determination of left ventricular mass in man. Anatomic validation of the method. Circulation 1977; 55: 613–618.
- Lapu-Bula R, Quarshie A, Lyn D, Oduwole A, Pack C, Morgan J, et al. The 894T allele of endothelial nitric oxide synthase gene is related to left ventricular mass in African Americans with high-normal blood pressure. J Natl Med Assoc 2005; 97: 197–205.
- 27. Kappagoda CT, Linden RJ, Newell JP. A comparison of the

oxygen consumption/body weight relationship obtained during submaximal exercise on a bicycle ergometer and on a treadmill. Quart J Exper Physiol Cognate Med Sci 1979; 64: 205–215.

- Jullien H, Ahmaidi S, Doutrellot PL, Telliez F, Bquferrache B, Libert JP, et al. Relationship between oxygen consumption and body mass during treadmill and cycle ergometry respectively. Sports Med Train Rehabil 1999; 9: 89–99.
- Franklin BA, Brinks J, Berra K, Lavie CJ, Gordon NF, Sperling LS. Using metabolic equivalents in clinical practice. Am J Cardiol 2018; 121: 382–387.
- Weidemann F, Niemann M, Breunig F, Herrmann S, Beer M, Stork S, et al. Long-term effects of enzyme replacement therapy on fabry cardiomyopathy: evidence for a better outcome with early treatment. Circulation 2009; 119: 524–529.
- Madsen CV, Bundgaard H, Rasmussen AK, Sorensen SS, Petersen JH, Kober L, et al. Echocardiographic and clinical findings in patients with Fabry disease during long-term enzyme replacement therapy: a nationwide Danish cohort study. Scand Cardiovasc J 2017; 51: 207–216.
- 32. El Dib R, Gomaa H, Ortiz A, Politei J, Kapoor A, Barreto F. Enzyme replacement therapy for Anderson-Fabry disease: a complementary overview of a Cochrane publication through a linear regression and a pooled analysis of proportions from cohort studies. PloS One 2017; 12: e0173358.
- Kubo T. Fabry disease and its cardiac involvement. J Gen Family Med 2017; 18: 225–229.
- Meehan SM, Junsanto T, Rydel JJ, Desnick RJ. Fabry disease: renal involvement limited to podocyte pathology and proteinuria in a septuagenarian cardiac variant. Pathologic and therapeutic implications. Am J Kidney Dis 2004; 43: 164–171.
- Hsu TR, Hung SC, Chang FP, Yu WC, Sung SH, Hsu CL, et al. Later onset Fabry disease, cardiac damage progress in

silence: experience with a highly prevalent mutation. J Am Coll Cardiol 2016; 68: 2554–2563.

- 36. Ramos-Jiménez A, Hernández-Torres RP, Torres-Durán PV, Romero-Gonzalez J, Mascher D, Posadas-Romero C, et al. The respiratory exchange ratio is associated with fitness indicators both in trained and untrained men: a possible application for people with reduced exercise tolerance. Clin Med Circ Respirat Pulm Med 2008; 2: 1–9.
- 37. Rostagno C, Olivo G, Comeglio M, Boddi V, Banchelli M, Galanti G, et al. Prognostic value of 6-minute walk corridor test in patients with mild to moderate heart failure: comparison with other methods of functional evaluation. Eur J Heart Fail 2003; 5: 247–252.
- Kim C, Choi HE, Lee KH, Kim YJ, Lee SJ. Influence of low peak respiratory exchange ratio on cardiac rehabilitation in patients with coronary artery disease. Ann Rehabil Med 2016; 40: 1114–1123.
- 39. Lamont LS. Gender differences in amino acid use during endurance exercise. Nutrit Rev 2005; 63: 419–422.
- 40. Numao S, Hayashi Y, Katayama Y, Matsuo T, Tanaka K. Sex differences in substrate oxidation during aerobic exercise in obese men and postmenopausal obese women. Metabolism 2009; 58: 1312–1319.
- Powell AW, Jefferies JL, Hopkin RJ, Mays WA, Goa Z, Chin C. Cardiopulmonary fitness assessment on maximal and submaximal exercise testing in patients with Fabry disease. Am J Med Genet Part A 2018; 176: 1852–1857.
- 42. Schmitz B, Thorwesten L, Lenders M, Duning T, Stypmann J, Brand E, et al. Physical exercise in patients with Fabry disease – a pilot study. Int J Sports Med 2016; 37: 1066–1072.
- Echevarria L, Benistan K, Toussaint A, Dubourg O, Hagege AA, Eladari D, et al. X-chromosome inactivation in female patients with Fabry disease. Clin Genet 2016; 89: 44–54.

